Author's response to reviews

Title: In-vitro evaluation of selected Egyptian traditional herbal medicines for treatment of Alzheimer disease

Authors:

Shereen K. Ali (sheryali57@yahoo.com)
Ahmed R. Hamed (n1ragab2004@yahoo.com)
Maha M. Soltan (mahasoltan@netscape.com)
Usama M. Hegazy (usama.hegazy@gmail.com)
Esameldin E. Elgorash (elgorashie@yahoo.com)
Ibrahim A. El-Garf (Ibrahim.elgarf@yahoo.com)
Ahmed A. Hussein (ahmohammed@uwc.ac.za)

Version: 3 Date: 13 April 2013

Author's response to reviews: see over
13-April-2013

**Journal**: BMC Complementary and Alternative Medicine  
**Manuscript ID**: 1851507638835080  
**Title**: In-vitro evaluation of selected Egyptian traditional herbal medicines for treatment of Alzheimer diseases  

Editorial comments:  
While we appreciate that improvements have been made to the language of your manuscript, we still feel that it would benefit from some further corrections to the style and grammar of written English being made. We would therefore recommend that you ask a native English speaking colleague to help you copyedit the paper.

*The article was corrected (two times) by native English speaker, which revised and improved the English as seen in the attached version, especially table 1.*

I hope that with the “English” corrections made, the paper will be accepted for publication.

We thank the editor for his help and guidance to improve the manuscript, and also for the careful revision.

Yours sincerely,

Ahmed A. Hussein